Author's response to reviews

Title: Takayasu's arteritis associated with Crohn's disease: a case report

Authors:

Nasser E Daryani (nebrahim@sina.tums.ac.ir)
Atoosa Nayer Habibi (anayery70@yahoo.com)
Mohammad Bashashati (bashashati_md@yahoo.com)
Mohammad Reza Keramati (dr_morezak@yahoo.com)
Moezedin Javad Rafiee (ajdarkosh1345@yahoo.com)
Hossein Ajdarkosh (ajdarkosh1345@yahoo.com)
Mehrdad Azmi (mehrdad_azmi@hotmail.com)

Version: 7 Date: 19 January 2008

Author's response to reviews:

Dear the BioMed Central Editorial Production Team,

Sorry for the delay. Regarding your recommended changes, we:
1- Added the email addresses of the authors for whom that had.
2- Edited the abstract with dividing it to introduction, case presentation, and conclusions.
3- Edited the references with adding the name of all authors.
4- Added the legend of the figures.

Sincerely yours,
Atoosa Nayer Habibi